Cruise 2010-13:  Line P June 2010
CTD/Rosette:
50 actual ‘casts’, but 53 files, 0001 to 0080.
0 stand-alone CTD casts.
0 ‘CTD’ casts with no sample taken (although there are some HEX files with no corresponding BTL files when the computer froze)
All casts done with deck unit 0424, CTD 0506.
All casts done with LARS (mid-ship station).
Seasave version 7.16
GENERAL NOTES:
Standard procedure:

The rosette was brought to the surface.  Pumps were turned ON.  Then the rosette was brought down to 10m and back up.  Data started to be archived then, with the rosette at the surface for a short time longer.  Then the cast would start. 

Cast 0080 was still archiving with the pumps on when the rosette got out of the water to (maybe?) help with figuring out Pressure offset. 
Problem casts:
Casts 0022/0023: same station, computer crash.  Cast 0022 has first two Niskins (1000 and 900 dbar).  Cast 0023 has the rest of the Niskins. 
Casts 0025/0026: same station, computer crashed on downcast.  All Niskins are in cast 0026.
Casts 0035/0036: same station, computer crashed on downcast.  All Niskins are in cast 0036.

Out of order (Table Driven) rosette casts:
Casts 0069, 0071. 
Casts 0070, 0075, 0076, 0078, 0079 were ‘Table Driven’ just to avoid using Niskin 1.

Headers:
Cast 0002 has “2010-12” as cruise number and no depth entered, should be 137. 
Cast 0003 has “2010-12 2010-13” as cruise number and no depth entered, should be 117.

Casts 0005, 0006, 0009, 0012, 0013, have “2010-12 2010-13” as cruise number.

Cast 0014: wrong station name, P4 should be P5. 

Cast 0016 has no depth entered, should be 2500.

Casts 0051 and 0052: space between P and 26.

Cast 0054: should be station PA-004, not P26. 

Cast 0070: should be station PA-003, not P26.

Cast 0071: should be called cast 0072. 

PAR and pH
Casts with PAR and pH sensors:  0001, 0002, 0003, 0005, 0013, 0022/0023, 0037, 0043, 0047, 0048, 0054, 0063, 0065, 0065, 0070
pH sensor only: 0012.
Cast 0047 and 0070: cap left on pH sensor. 

Niskins at CTD/Rosette casts:
Cast 0001: 11 Niskins for the cast, the others closed at the surface to check the rosette. 
Casts 0019, 0020: 2 Niskins closed at CHL max, 2 Niskins closed at 5m, samples out of Niskins 1 and 3.
Cast 0027: 17 Niskins closed, but the three at 100 dbar (12-14) are there by mistake … so not sampled and no sample number.  They can simply be removed from the CHE file.
Cast 0028: a few Oxygen samples taken to calibrate the Argo float, plus the regular Chl Max and 5 m samples. 
Casts 0005, 0009, 0012, 0018, 0030, 0033, 0039, 0043: Niskin 1 did not close. 

Cast 0018: Niskin 3 didn’t close either. 
Cast 0041: sample number 251: mistrip.
Loops:
Loops were taken at almost every station for Oxygen, Nutrients, Chlorophyll and Salinity.  Only a few “major stations” don’t have loops.
TSG: 

The TSG seems to be working good this time! 
Sample analysis: 
Dissolved oxygen samples were analysed on board the ship.

Chlorophyll samples were analysed on board the ship.

Nutrients samples were brought back and will be analysed in the lab.

Salinity samples for P8 and P16 were analysed on board the ship on the Portasal. All other salinity samples will be analysed on the Autosal at IOS. 
